Author’s response to reviews

Title: Prioritization strategies for pandemic influenza vaccine in 27 countries of the European Union and the Global Health Security Action Group: a review

Authors:

Masja Straetemans (StraetemansM@rki.de)
Udo Buchholz (BuchholzU@rki.de)
Sabine Reiter (ReiterS@rki.de)
Walter Haas (HaasW@rki.de)
Gerard Krause (KrauseG@rki.de)

Version: 5 Date: 6 September 2007

Author’s response to reviews: see over
6 September 2007
Editor
BMC Public Health

Dear Editor,

We are submitting our fourth revision of the manuscript titled, “Prioritization strategies for pandemic influenza vaccine in 27 countries of the European Union and the Global Health Security Action Group: a review”.

We have deleted the headers related to table 2,4,5 continued. After discussion with editorial services in which we motivated that it was impossible to change all tables in portrait-format, we have been told that the current format of the tables would be ok.

We hope we have been able to revise the manuscript including the formatting changes to your satisfaction.

Sincerely,

Gérard Krause
Robert Koch-Institute
Department of Infectious Disease Epidemiology
Seestrasse 10, 13352 Berlin, Germany
Phone: +49 (0) 30 18754 3404
Fax: +49 (0) 30 18754 3533
Email: krauseg@rki.de